Author's response to reviews

Title: Trends In The Histopathology Of Childhood Nephrotic Syndrome In Ibadan Nigeria: Preponderance Of Idiopathic Focal Segmental Glomerulosclerosis

Authors:

Adanze O Asinobi (adasinobi@yahoo.com)
Adebowale D Ademola (dr_deboademola@yahoo.co.uk)
Clement A Okolo (clemabuokolo@yahoo.ca)
Joseph O Yaria (jyaria@yahoo.com)

Version: 6 Date: 1 November 2015

Author's response to reviews: see over
Dear Sir,

COVERING LETTER

Thank you for your mail.

The age of assent was Seven years and it has been highlighted in red on line 73.

The Former Table II has been deleted, and the figures have also been deleted from the manuscript.

We have carried out a further grammar and spelling editing and a few changes have been highlighted in red.

Thank you and best regards.

Yours faithfully,

Dr. Adanze O. Asinobi.